Author's response to reviews

Title: Liquid and vapour-phase antifungal activities of selected essential oils against Candida albicans: Microscopic observations and chemical characterization of Cymbopogon citratus

Authors:

Amit K Tyagi (amittyagiiitd@gmail.com)
Anushree Malik (anushree_malik@yahoo.com)

Version: 5 Date: 8 November 2010

Author's response to reviews: see over
Dear Editor,

Please find herewith the response to “Formatting Changes” suggested on the above-referred manuscript. The required changes have been made and the revised manuscript is being uploaded.

Kind regards
Amit Tyagi

--------------------------------------------------------------------------------------------------

Formatting changes to be made:
-------------------------------

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

Done

Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript.

Done

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Materials and methods: please rename this section 'Methods'.

Done

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Done

Captions: please rename this section 'Figure legends'.

Done

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

Done
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Thank you
-----------------------------------


To submit your revised manuscript
-----------------------------------

1. Go to 'my BioMed Central' (http://www.biomedcentral.com/my/) and log on with your email address and password. Then click on 'My manuscripts'. You will reach a page giving details of all the manuscripts you have submitted to journals published by BioMed Central.

2. Click on the 'Submit revision' button next to the title of this manuscript.

3. On the 'Manuscript details' tab, please update the title, abstract and author details if they have changed since the previous version.

4. If the two sources are different after resubmission, we will take the information in the manuscript to be the most recent and hence the most accurate and use it upon publication.

5. On the 'Cover letter' page, please provide a covering letter with a point-by-point description of the changes made.

6. On the 'Upload files' page, please upload the revised version of the manuscript and click 'Submit new version'. Please wait for the confirmation page to appear - this may take a few moments.

Please contact us if you have any problems or questions.

Best wishes,